Author’s response to reviews

Title: The COMT-polymorphism is not associated with the incidence of acute kidney injury after cardiac surgery - a prospective cohort study

Authors:

Mehmet Oezkur (mehmet.oezkur@uk-halle.de)
Attila Magyar (Magyar_A@ukw.de)
Phillip Thomas (E_Thomas_P@ukw.de)
Stefan Störk (Stoerk_S@ukw.de)
Peter U. Heuschmann (E_Heuschma_P@ukw.de)
Rainer G. Leyh (Leyh_r@ukw.de)
Martin Wagner (Wagner_M@ukw.de)

Version: 3 Date: 21 Dec 2017

Author’s response to reviews:

Dear editors,

we thank you for accepting our draft. The minor revisions were addressed and we hope that now the draft is suitable for publication.

1. Abstract:
The abstract was changed according to the comments.

2. Headings:
The headings were changed according to the comments.

3. Abbreviations:
The abbreviation list was added.

4. Declarations:
A declaration section was added.
5. The reference for table 3 was taken out.

Best Regards

Mehmet Oezkur